Author’s response to reviews

Title: Intra and post-operative complications observed with femtosecond laser-assisted cataract surgery versus conventional phacoemulsification surgery: A systematic review and meta-analysis

Authors:

Jinhua Wang (oculist498474@163.com)
Fanfan Su (64964747@qq.com)
Yong Wang (395300060@qq.com)
Yao Chen (287740@qq.com)
Qiao Chen (50231474@qq.com)
Fen Li (565227656@qq.com)

Version: 2 Date: 13 Jun 2019

Author’s response to reviews:

POINT TO POINT RESPONSE TO EDITORS

Editor Comments:

1. Author Contributions

-- Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

AUTHORS’ RESPONSE: THANK YOU. APPROPRIATE CHANGES WERE MADE AS SUGGESTED.

2. Additional Files

-- Thank you for providing your PRISMA checklist and flow diagram, and track changes as additional files. As these are no longer required, please remove them when submitting the revised version of your manuscript.

AUTHORS’ RESPONSE: OK. WE HAVE REMOVED THEM APPROPRIATELY.

3. Clean Manuscript
-- At this stage, we ask that you submit a clean version of your manuscript and do not include track changes or highlighting.

AUTHORS’ RESPONSE: THANK YOU. OK.